Author's response to reviews

Title: Effect of spinal manipulation on sensorimotor functions in back pain patients: study protocol for a randomised controlled trial

Authors:

David G Wilder (david-wilder@uiowa.edu)
Robert D Vining (robert.vining@palmer.edu)
Katherine A Pohlman (katherine.pohlman@palmer.edu)
William C Meeker (meeker_b@palmer.edu)
Ting Xia (ting.xia@palmer.edu)
James W DeVocht (james.devocht@palmer.edu)
R. Maruti Gudavalli (ram.gudavalli@palmer.edu)
Cynthia R Long (long_c@palmer.edu)
Edward F Owens (eowens@nwhealth.edu)
Christine M Goertz (christine.goertz@palmer.edu)

Version: 4 Date: 23 June 2011

Author's response to reviews:

Editorial Production Team-

Thank you for your response regarding the change of tense. We have decided to keep in the tense that was originally accepted. All the changes you requested have been made, as well as some editorial changes beyond those. Additional, we updated an authors affiliation and e-mail address (Ed Owens).

Regards,

Christine Goertz